ss22-May-03
Week 21: (26/05 to 30/05)

SM18 Operation SCHEDULE (Monday Morning) Meeting 

· Test Coordinator this week:    V.Chohan

· MTA Technician on call:  (16-3602) G.Deferne.

1. Planning for this week : 
· TBA1:   1022  :  End of the test at 1.9 K for 1022
· TBA2:   warm up ,Tests after final warm up: 1016 is coming.
· TBC2 :  3013 at 1.9 K (second run after thermal cycle) and warm up. 
·  TBC1 : 1024 at 4.2 K tested when 3013 is finished.
· *** CONTINUE  USAGE of “ To-do list ”    *** 
2.  Problems from Operation Last week ( Olav+ Giorgio)
· Problems with TRU 1 and shaft 25: (Connectors )

· TBA1: One amplifier card was missing in the mobile rack. 

· Software problems MMP: MXI errors : We quit and re-start MMP.

· Bench C2: The acquisition received no trigger for aperture 1 before starting the full load line. MMP was closed and re-launched. 

· Cluster C2: The test “degaussing” was not implemented: It was implemented for C1. Problem resolved.

· Quench at 1350 mbar at 2 A/s: temperature not stable for 3013, no temperature sensor recorded for the 2006.

· Water problems: not enough water to run three power supplies. Water closed on TBC to run the tests on TBA.

· ….

3. Situation From Cryogenics:  J-P Lamboy
4. Other General Issues.

· Additional checks for instrumentation at warm after the final warm up (P.Pugnat)

· Courses in SM18 (string control room): O.Berrig

